Author's response to reviews

Title: Two case reports of bilateral vertebral artery tortuosity and spiral twisting in vascular vertigo

Authors:

zhang hongtao (zhanghongtao.13526676180@yahoo.com.cn)
zhang shuling (zhangshuling99@yahoo.com.cn)
zhang daopei (zhangdaopei89@163.com)

Version: 2 Date: 8 May 2013

Author's response to reviews: see over
1. Consent to publish the report and to publish the images, from each patient.

*We have used the following wording for the consent section: "Written informed consent was obtained from the patient for publication of this case report and any accompanying images. A copy of the written consent is available for review by the Editor-in-Chief of this journal."*

2. Authors’ information:

*We have placed the Authors’ Contributions section after Competing interests.*

3. Competing interests:

*No competing interests for our paper, we have placed a declaration of competing interests in our manuscript.*

4. Author groups:

*We have added the name to the 'first name' box; and the bracketed abbreviation to the 'last name' box in the submission system. We also removed the names of the members of the group from the title page and place them in the Acknowledgements.*

5. Please make general conclusions based on these two cases.

*We have amended our conclusion.*

6. Please remove the duplicate figures from the main manuscript.

*We have removed the duplicate figures from the main manuscript.*